Author’s response to reviews

Title: Proliferation and survival molecules implicated in the inhibition of BRAF pathway in thyroid cancer cells harbouring different genetic mutations.

Authors:

Ana Preto (apreto@ipatimup.pt)
Joana Gonçalves (jgoncalves@cnc.cj.uc.pt)
Ana P Rebocho (arebocho@ipatimup.pt)
Joana Figueiredo (jfigueiredo@ipatimup.pt)
Ana M Meireles (ameireles@ipatimup.pt)
Ana S Rocha (arocha@ipatimup.pt)
Helena M Vasconcelos (hvasconcelos@ipatimup.pt)
Hugo Seca (hteixeira@ipatimup.pt)
Raquel Seruca (rseruca@ipatimup.pt)
Paula Soares (psoares@ipatimup.pt)
Manuel Sobrinho-Simões (ssimoes@ipatimup.pt)

Version: 2 Date: 16 February 2009

Author’s response to reviews:

Porto, 16th of February of 2009

To:

Professor Melissa Norton
Editor-in-Chief of BMC Cancer

Dear Professor Melissa Norton,

Please find attached a revised version of our manuscript that we would like to resubmit to BMC Cancer, “Proliferation and survival molecules implicated in the inhibition of BRAF pathway in thyroid cancer cells harbouring different genetic mutations”.

Following your suggestions we have removed all the data concerning NPA cell line, remade all the figures and revised the manuscript in accordance, not changing the general scope of the manuscript. We have also added two recent and relevant new references (number 22 and 36).

We hope that the manuscript is now suitable for consideration to be published in BMC Cancer.

Looking forward to receive your news, sincerely yours,